Author’s response to reviews

Title: Prevalence of lymphedema in women under treatment for breast cancer in a referral center in southeastern Brazil

Authors:

Daniella MF Paiva (jcndani@yahoo.com.br)
Vivian O Rodrigues (viviolliver@hotmail.com)
Marcelle G Cesca (marcelle.goldner@gmail.com)
Pamella V Palma (pamellavalente@hotmail.com)
Isabel CG Leite (icgleite@hotmail.com)

Version: 6 Date: 22 December 2012

Author’s response to reviews:

In response to the reviewer Ms Wella Valenzuela,

The quality of the written English was improved by a professional language editing service, Edanz, as BioMed Central suggested.

We are at disposal.

The authors